Author’s response to reviews

Title: The role of oxytocin receptor gene (OXTR) DNA methylation (DNAm) in human social and emotional behaviour: a systematic narrative review.

Authors:

Catherine Maud (maud_catherine@hotmail.com)
Joanne Ryan (joanne.ryan@mcri.edu.au)
Jennifer McIntosh (mcintosh@familytransitions.com.au)
Craig Olsson (Craig.Olsson@rch.org.au)

Version: 2 Date: 29 Apr 2018

Author’s response to reviews:

Reviewers reports’ response: BPSY-D-17-00920 R1 (re-submission)

(Previously responded to reviewers 1-4 from submission BPSY-D-17-00527/ BPSY-D-17-01080 with previous submission)

BMC Psychiatry
Editorial Office

29 April 2018

Dear Catherine Olino

Thank you for forwarding the final recommendations from both reviewers to accept our manuscript for publication.

Reviewer 1 required no further clarification of issues and to accept without revision.

Reviewer 2 submitted an incorrect version of his review, which on clarification from the Editorial Office, were consistent with Reviewer 1 in requiring no further revisions.
In light of these comments, we attach a final version of our manuscript, with all track-changes and other edits removed for your convenience. We look forward to hearing the final decision of your office in due course.

Thank you again for your support through the review process.

Kind regards,

Dr Catherine Maud

Deakin University, Faculty of Health, School of Psychology, Centre for Social and Early Emotional Development (SEED), Geelong and Burwood, Victoria 3125 Australia